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ABSTRACT 

The global concern for fossil fuel depletion and emission of greenhouse gases has led to 

increased research on the production of transportation fuels, alternate fuels, and value-added 

chemicals from the renewable 2nd generation lignocellulosic biomass. Among the different 

routes, fast pyrolysis is an eco-friendly, efficient, and cost-effective technology for biomass 

conversion, which produces liquid bio-oil (~80% yield), solid biochar, and pyrolytic gases. 

However, immediate commercial implementation of the condensed bio-oil formed by 

pyrolysis is impossible due to the presence of a high percentage of oxygen functionalities 

(~30-40%). The existing upgrading process includes the catalytic hydrodeoxygenation (HDO) 

of bio-oil (to remove oxygenates); however, it operates under harsh reaction conditions, which 

results in the loss of C1‒C4 light oxygenate fractions (5-15%) in the form of non-condensable 

gases that could have otherwise contributed in increasing the heating value of the biofuel and 

chemicals. This loss can be suppressed by valorization of lignin-derived phenolic compounds 

via C–C coupling by an alkylation reaction with such light oxygenates to produce C10–C13 

fuel range alkylphenols prior to HDO. The present work is focused on the alkylation of m-

cresol, an abundant species in pyrolysis bio-oil, with iso-propanol (as light oxygenate) over 

metal loaded zeolites to form stable C10 fuel precursor and important value-added chemical, 

thymol. 

A thermodynamic analysis is performed to understand the thermodynamic limits of 

the m-cresol alkylation reaction with iso-propanol alongwith all the possible side-reactions to 

assess the thermodynamically controlled and kinetically controlled reactions. After 

identifying the thermodynamic boundaries of this reaction, the zeolites HZSM5, HMCM22, 

HBEA, and HY having different medium and large pore architecture and varied acidity are 

screened as catalysts for maximizing the desired mono-C-alkylated product, thymol, in a 

continuous flow atmospheric packed bed reactor. The structure-activity relationship is found 



 

v 
 

to be highly significant for this reaction. Therefore, the tuning of Bronsted and Lewis acidity 

and pore dimensions of the zeolite catalysts by ion exchanging with different metals are 

performed and the amount of metal is optimized, which enhances the conversion of m-cresol 

to near equilibrium conversion and with the selective formation of thymol. The disruption in 

structure and stability of the zeolite framework and its degraded catalytic performance is also 

observed at a very high amount of metal loading. After choosing the promising catalyst, 

statistical design of experiments is used to investigate the effect of important process 

parameters like reaction temperature, weight hourly space velocity, iso-propanol to m-cresol 

mole ratio, and their individual and interactive effects on the conversion of m-cresol and 

thymol selectivity in alkylation reaction for optimization of process parameters over the most 

promising catalyst. 

Mechanistic insights into this reaction mechanism over metal loaded zeolite are 

investigated using Density Functional Theory (DFT) calculations and the effect of acidity 

on the reaction mechanism is elucidated by comparing the results with the parent zeolite. 

The rate of reaction and thermodynamic stability of thymol is found to be higher in case 

of metal loaded large pore protonated zeolite, resulting from higher acidity of this catalyst. 

Finally, the kinetics of this reaction is investigated over the most promising catalyst in an 

external and intraparticle mass diffusion-free regime. The time-on-stream behavior shows that 

the catalyst deactivates, especially at high temperatures (~300 °C) due to active site loss and 

pore plugging by coke. However, the rate of deactivation of this modified catalyst is 

significantly lower than the rate of deactivation of the parent zeolite. Based on the product 

distribution, a reaction network is elucidated and a kinetic model is developed. The kinetic 

parameters are estimated in the temperature range of 200 °C to 300 °C. The catalyst 

deactivation is also included in the model to check the temperature dependence of the catalyst 

deactivation parameters. 



सार 

जीवाश्म ईंधन की कमी और ग्रीनहाउस गैस  ंके उत्सजजन के लिए वैलिक ल ंता ने अक्षय लितीय पीढी के 

लिग्न सेिुि लसक बाय मास से पररवहन ईंधन, वैकल्पिक ईंधन, और मूल्य वलधजत रसायन  ंके उत्पादन पर 

अनुसंधान क  बढा लदया है। लवलिन्न मागों में, तेजी से पायर लिलसस बाय मास रूपांतरण के लिए एक 

पयाजवरण-अनुकूि, कुशि और िागत प्रिावी तकनीक है, ज  तरि जैव-तेि (~ 80% उपज), ठ स 

बाय  ार और पाइर िाइलिक गैस  ंका उत्पादन करती है। हािांलक, पायर लिलसस िारा गलठत संघलनत 

जैव-तेि का तत्काि वालणल्पिक कायाजन्वयन ऑक्सीजन फंकं्शलसिीज़ (~ 30-40%) के उच्च प्रलतशत की 

उपल्पथिलत के कारण असंिव है। मौजूदा अपगे्रल ंग प्रलिया में बाय -ऑइि के उते्प्ररक 

हाइ र   ेक्सीजनेशन (ए  ीओ) (ऑक्सीजन क  हिाने के लिए) शालमि हैं; हािााँलक, यह कठ र प्रलतलिया 

पररल्पथिलतय  ंमें काम करता है, लजसके पररणामस्वरूप C1 light C4 प्रकाश ऑक्सीकारक अंश  ं(5-15%) 

की हालन ह ती है, ज  गैर-संघनन य ग्य गैस  ंके रूप में ह ती है ज  जैव ईंधन और रसायन  ंके ताप मूल्य 

क  बढाने में य गदान दे सकती िी।ं । इस नुकसान क  लिलग्नन-वु्यत्पन्न फेन लिक यौलगक  ंके सत्यापन 

िारा दबाया जा सकता है, सी-सी युग्मन के माध्यम से एक क्षारीय प्रलतलिया के साि इस तरह के प्रकाश 

ऑक्सीजन के साि सी 10-सी 13 ईंधन रेंज एल्काइफेन ल्स का उत्पादन करने के लिए ए  ीओ से पहिे। 

वतजमान कायज m-cresol के एले्किाइजेशन पर कें लित है, पाइर लिलसस जैव-तेि में एक प्र ुर मात्रा में, 

आइस -प्र पेन ि (प्रकाश ऑक्सीिेज के रूप में) धातु ि   लकए गए लजओिाइि्स के साि ल्पथिर C10 

अग्रदूत और महत्वपूणज मूल्य वलधजत रसायन, िाइम ि बनाने के लिए है। 

िमो ायनालमक रूप से लनयंलत्रत और काइनेलिक लनयंलत्रत प्रलतलियाओ ंका आकिन करने के लिए सिी 

संिालवत दुष्प्रिाव  ं के साि-साि आइस -प्र पेन ि के साि एम-िेस ि अल्काइिेशन प्रलतलिया की 

िमो ायनालमक सीमाओ ं क  समझने के लिए एक िमो ायनालमक लवशे्लषण लकया जाता है। इस 

प्रलतलिया की िमो ायनालमक सीमाओ ंकी पह ान करने के बाद, लजओिाइि्स HZSM5, HMCM22, 



HBEA, और HY के पास अिग-अिग मध्यम और बडे ताकना वासु्तकिा और लवलवध अम्लता है ज  

वांलित म न -सी-अल्कीिेिे  उत्पाद, िाइम ि क  अलधकतम प्रवाह में अलधकतम करने के लिए उते्प्ररक 

के रूप में जां ा जाता है। वायुमं िीय पैक लबस्तर ररएक्टर। इस प्रलतलिया के लिए संर ना-गलतलवलध 

संबंध अत्यलधक महत्वपूणज पाया जाता है। इसलिए, लवलिन्न धातुओ ंके साि आयन एक्स ेंज िारा ब् सें्ट  

और िुईस अम्लता और लजओिाइि उते्प्ररक के ताकना आयाम  ंकी टू्यलनंग की जाती है और धातु की 

मात्रा क  अनुकूलित लकया जाता है, ज  सम-लवषम रूपांतरण के साि m-cresol के रूपांतरण क  बढाता 

है और  यनात्मक गठन के साि ह ता है। अजवाइन का सत्व। लजओिाइि ढां े की संर ना और ल्पथिरता 

में व्यवधान और इसके अपमालनत उते्प्ररक प्रदशजन क  धातु ि ल ंग की बहुत अलधक मात्रा में मनाया 

जाता है। ह नहार उते्प्ररक क   ुनने के बाद, प्रय ग  ं के सांल्पिकीय ल जाइन का उपय ग प्रलतलिया 

तापमान, वजन प्रलत घंिा अंतररक्ष वेग, आइस -प्र पेनॉि से िेकर एम-िेस ि लति अनुपात जैसे महत्वपूणज 

प्रलिया मापदं   ंके प्रिाव की जां  करने के लिए लकया जाता है, और एम के रूपांतरण पर उनके 

व्यल्पिगत और इंिरैल्पक्टव प्रिाव। सबसे ह नहार उते्प्ररक पर प्रलिया मापदं   ंके अनुकूिन के लिए 

क्षारीय प्रलतलिया में िेक ि और िाइम ि  यनात्मकता। 

मेिि ि  े  ज़ िाइि पर इस प्रलतलिया तंत्र में यंत्रवत अंतर्दजलि की जााँ   ेंलसिी फंक्शनि थ्य री 

( ीएफिी) गणनाओ ं का उपय ग करके की जाती है और प्रलतलिया तंत्र पर अम्लता के प्रिाव क  

अलििावक लजओिाइि के साि पररणाम  ंकी तुिना करके स्पि लकया जाता है। िाइम ि की प्रलतलिया 

और िमो ायनेलमक ल्पथिरता की दर इस उते्प्ररक की उच्च अम्लता के पररणामस्वरूप धातु ि   लकए 

गए बडे लिियुि ज़ िाइि के मामिे में अलधक पाई जाती है। अंत में, इस प्रलतलिया के कैनेिीक्स की 

जां  एक बाहरी और अंतगजिाजशयी जन प्रसार-मुि शासन में सबसे आशाजनक उते्प्ररक से अलधक की 

जाती है। समय पर धारा व्यवहार से पता  िता है लक उते्प्ररक सलिय साइि के नुकसान के कारण लवशेष 

रूप से उच्च तापमान (~ 300 ल ग्री सेल्पल्सयस) पर लनल्पिय ह  जाता है और क क िारा लििण करता है। 

हािांलक, इस संश लधत उते्प्ररक क  लनल्पिय करने की दर माता-लपता लजओिाइि के लनल्पिय ह ने की 



दर से काफी कम है। उत्पाद लवतरण के आधार पर, एक प्रलतलिया नेिवकज  क  स्पि लकया जाता है और 

एक गलतज मॉ ि लवकलसत लकया जाता है। गलतज मापदं   ं क  200 ल ग्री सेल्पल्सयस से 300 ल ग्री 

सेल्पल्सयस के तापमान रेंज में अनुमालनत लकया जाता है। उते्प्ररक लनल्पियकरण मापदं   ंके तापमान 

लनिजरता की जां  करने के लिए उते्प्ररक लनल्पियकरण क  िी मॉ ि में शालमि लकया गया है। 
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